Author's response to reviews

Title: Assessment of the effect of iris colour and having children on 5-year risk of death after diagnosis of uveal melanoma: a follow-up study.

Authors:

Andrea Schmidt-Pokrzywniak (andrea.schmidt-pokrzywniak@uk-halle.de)
Sven Kalbitz (sven.kalbitz@student.uni-halle.de)
Oliver Kuss (oliver.kuss@medizin.uni-halle.de)
Karl-Heinz Jöckel (k-h.joeckel@uk-essen.de)
Norbert Bornfeld (retina@uni-essen.de)
Andreas Stang (andrea.stang@uk-halle.de)

Version: 2 Date: 7 October 2013

Author's response to reviews: see over
Dear Editor,

Enclosed please find our received manuscript entitled “Assessment of the effect of iris colour and having children on 5-year risk of death after diagnosis of uveal melanoma: a follow-up study.”.

Below please find point by point our reactions to the reviewers’ comments. If there is anything we can do to facilitate the Journal’s consideration of our paper, please do not hesitate to let me know.

With kind regards

Andrea Schmidt-Pokrzywniak on the behalf of all authors

Author for correspondence
Andrea Schmidt-Pokrzywniak M.A., PhD;
Institute of Clinical Epidemiology
Medical Faculty, University of Halle-Wittenberg, 06097 Halle, Magdeburgerstr. 8, Germany
Phone: +49-345-557-4469       Fax: +49-345-557-3565
e-mail: andrea.schmidt-pokrzywniak@uk-halle.de

(1) Acknowledgements
Please acknowledge anyone who contributed towards the article by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship.

There was no person other than the authors

(2) Requesting ethics statement and (3) Requesting consent statement
I have add the statements in the methods section